Author's response to reviews

Title: Translating research into maternal health care policy: a qualitative case study of the use of evidence in policies for the treatment of eclampsia and pre-eclampsia in South Africa

Authors:

Karen Daniels (karen.daniels@mrc.ac.za)
Simon Lewin (simon.lewin@nokc.no)
Practihc Policy Group (PPG) (not@valid.com)

Version: 9 Date: 8 December 2008

Author's response to reviews:

The names of the Practihc Policy Group got left out of the last version you sent. I have included these again along with the email addresses. We need to keep these names in the document as this was part of our agreement as a group. Keeping each of their email addresses is not essential. Originally we listed the names of the members of the group under the author contributions. Doing so again could work too if you don't want all the names at the beginning of the paper.